Author’s response to reviews

Title: Absorbable suture can be effectively and safely used to close the mesenteric defect in a Gastric bypass Sprague-Dawley Rat Model

Authors:
Libin Yao (yaolibin_123@126.com)
Ponnie Dolo (ponnie85@yahoo.com)
Yong Shao (shaoyutong1975@163.com)
Chao Li (18068718716@189.cn)
Jason Widjaja (drjwbs@yahoo.com)
Jian Hong (hongjiajian@qq.com)
Xiaocheng Zhu (zhuxccf@163.com)

Version: 2 Date: 21 Oct 2019

Author’s response to reviews:

Reviewer reports:
Sonja Chiappetta, MD (Reviewer 2): Congratulations for the improvement. Poorly, a step-by-step answer to the reviewer comments is missing.
Response: Thank you for your commendation. We did provided a step-by-step answer to the reviewer comments in a separate file (response to reviewer comments) submitted as supplimental material along with the manuscript.

The paper improved with the corrections, nevertheless there are a lot of spelling mistakes: f.e.
page 3 Stenber g et al
page 4 Even so, Most studies, reduce The risk of IH, mesenteric defect. , established a Rat model of
page 5 Performing the gastric bypass in obese rat
page 6 food intake was measure
page 8 All animal survive the surgery and follow-up for 8 weeks.
page 10 many materials are currently being by surgeons - verb missing!
page 11 Remarkably, our experiment results indicated that adhesion in prolene group was minimum.
Response: Thank you for your commendation. We apologize for the spelling mistakes. The corrections have been made in the manuscript accordingly.

The paper has to be revised by a native english speaker!
Response: The paper has been review by a native english speaker.